Craig Chambers CSE 501 (Winter ’96) Reading Assignments

Reading Assignment, Weeks #9 & 10

For: Read:

2/26 [Shivers 88] 2/28 3/1 [Stroustrup 89]

3/4 [Agesen & Hölzle 95], [Dean et al. 95] 3/6 3/8

Bibliography:

[Shivers 88] Olin Shivers. Control Flow Analysis in Scheme. In PLDI’88. [Stroustrup 89] Bjarne Stroustrup. Multiple Inheritance for C++. In Computing Systems 2(4), pp.

367-395, Fall, 1989.

[Agesen & Hölzle 95] Ole Agesen and Urs Hölzle. Type Feedback vs. Concrete Type Inference: A Comparison of Optimization Techniques for Object-Oriented Languages. In OOPSLA’95.

[Dean et al. 95] Jeffrey Dean, David Grove, and Craig Chambers. Optimization of Object-Oriented Programs Using Static Class Hierarchy Analysis. In ECOOP ’95.